Author's response to reviews

Title: A Changing Picture of Shigellosis in Ho Chi Minh City, Viet Nam; Shifting Species Dominance, Antimicrobial Susceptibility and Clinical Presentation

Authors:

Ha Vinh (vinhh@oucru.org)
Nguyen TK Nhu (nhunkt@oucru.org)
Tran VT Nga (ngatvt@oucru.org)
Pham T Duy (duypt@oucru.org)
James I Campbell (jcampbell@oucru.org)
Nguyen VM Hoang (hoangnvm@oucru.org)
Maciej F Boni (mboni@oucru.org)
Pham VT My (mypvt@oucru.org)
Christopher Parry (cmparry@liverpool.ac.uk)
Tran TT Nga (ngattt@oucru.org)
Pham V Minh (minhvp@oucru.org)
Coa T Thuy (thuyct@oucru.org)
To S Diep (tsdiep@oucru.org)
Le T Phoung (ltp@oucru.org)
Mai N Lanh (mainl@oucru.org)
Bui L Mong (monglbui@oucru.org)
Vo TC Anh (anhtcv@oucru.org)
Phan VB Bay (bayvbp@oucru.org)
Nguyen VV Chau (chaunvv@oucru.org)
Jeremy Farrar (jfarrar@oucru.org)
stephen baker (sbaker@oucru.org)

Version: 3 Date: 27 August 2009

Author's response to reviews:

Enclosed is a revised submission of the manuscript. It was considered unsuitable for submission to BMC microbiology, therefore I wish the manuscript to be considered for BMC infectious diseases.

The manuscript has been modified to account for the clinical trial information and a more in depth description of the studies used for data analysis. Neither study B or study C have been published previously, study B was not a clinical trial, therefore there is only limited clarification required. Study C trial number and other information about the trial have been included. I hope the article is now acceptable for potential publication.

Stephen Baker